Author's response to reviews

Title: An Approach to Measure Compliance to Clinical Guidelines in Psychiatric Care.

Authors:

Tord Forsner (tord.forsner@ki.se)
Anna Aberg Wistedt (anna.wistedt@sll.se)
Mats Brommels (mats.brommels@ki.se)
Yvonne Forsell (yvonne.forsell@ki.se)

Version: 7 Date: 8 July 2008

Author's response to reviews: see over
Response to comments from reviewer

Concerning the manuscript: An Approach to Measure Compliance to Clinical Guidelines in Psychiatric Care. MS:1300316886178997.

Dear Mr Kerr,

We thank the reviewer, James Rohrer for the valuable remarks. We have made the suggested revisions.

The reviewer suggested four models to compare the compliance score between the intervention and the control groups while adjusting for age and gender.

We have now performed the suggested analysis. We have chosen to use model one and two using multiple regression to compare the difference between the two measurements of compliance scores between the intervention and the control groups, adjusted for age and gender. This have now been added under the statistical analysis section in method, page 9 and in result section, end of first paragraph, page 10 and last of first paragraph, page 12.

We thanks the reviewer for the suggestion and agree that this model contribute to our result section.

We hope that the manuscript will now meet the requirements for publication. Once again we thank the reviewers for the valuable comments to our manuscript.

Stockholm the 7 of July 2008

With best wishes

Tord Forsner
Yvonne Forsell
Anna Åberg Wistedt